

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Foster-McGregor, Neil; Stehrer, Robert

Working Paper

Sectoral Productivity, Density and Agglomeration in the Wider Europe

wiiw Working Paper, No. 47

Provided in Cooperation with:

The Vienna Institute for International Economic Studies (wiiw) - Wiener Institut für Internationale Wirtschaftsvergleiche (wiiw)

Suggested Citation: Foster-McGregor, Neil; Stehrer, Robert (2008): Sectoral Productivity, Density and Agglomeration in the Wider Europe, wiiw Working Paper, No. 47, The Vienna Institute for International Economic Studies (wiiw), Vienna

This Version is available at: https://hdl.handle.net/10419/203912

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



The Vienna Institute for International Economic Studies

Working Papers | 47 | September 2008

Neil Foster and Robert Stehrer

Sectoral Productivity, Density and Agglomeration in the Wider Europe



wiiw Working Papers published since 1995:

- No. 8 J. Fersterer and M. Landesmann: Vertical Product Differentiation in International Trade: An Econometric Investigation. April 1998
- No. 9 A. Bhaduri, K. Laski and M. Riese: Effective Demand versus Profit Maximization in Aggregate Demand/Supply Analysis from a Dynamic Perspective. November 1998
- No. 10 M. Landesmann and R. Stehrer: The European Unemployment Problem: A Structural Approach. March 1999
- No. 11 R. Stehrer, M. Landesmann and J. Burgstaller: Convergence Patterns at the Industrial Level: the Dynamics of Comparative Advantage. October 1999
- No. 12 K. Laski: Three Ways to . . . High Unemployment. January 2000
- No. 13 F. Turnovec: A Leontief-type Model of Ownership Structures. Methodology and Implications. April 2000
- No. 14 M. Landesmann and R. Stehrer: Potential Switchovers in Comparative Advantage: Patterns of Industrial Convergence. June 2000
- No. 15 R. Stehrer: Industrial Specialization, Trade, and Labour Market Dynamics in a Multisectoral Model of Technological Progress. January 2001; revised version February 2002
- No. 16 K. Laski and R. Römisch: Growth and Savings in USA and Japan. July 2001
- No. 17 P. Egger and R. Stehrer: International Outsourcing and the Skill-Specific Wage Bill in Eastern Europe. July 2001
- No. 18 R. Stehrer: Expenditure Levels, Prices and Consumption Patterns in a Cross-Section of Countries. August 2001
- No. 19 R. Stehrer and J. Wörz: Technological Convergence and Trade Patterns. October 2001
- No. 20 M. Landesmann, R. Stehrer and S. Leitner: Trade Liberalization and Labour Markets: Perspective from OECD Economies. October 2001
- No. 21 E. Christie: Potential Trade in South-East Europe: A Gravity Model Approach. March 2002
- No. 22 M. Landesmann and R. Stehrer: Technical Change, Effective Demand and Economic Growth. April 2002
- No. 23 R. Stehrer and J. Wörz: Industrial Diversity, Trade Patterns and Productivity Convergence. November 2002; revised version July 2003
- No. 24 E. Christie: Foreign Direct investment in Southeast Europe: a Gravity Model Approach. March 2003
- No. 25 J. Wörz: Skill Intensity in Foreign Trade and Economic Growth. November 2003; revised version January 2004
- No. 26 M. Landesmann and R. Stehrer: Global Growth Processes: Technology Diffusion, Catching-up and Effective Demand. January 2004
- No. 27 M. Landesmann and R. Stehrer: Modelling International Economic Integration: Patterns of Catching-up, Foreign Direct Investment and Migration Flows. March 2004
- No. 28 L. Podkaminer: Assessing the Demand for Food in Europe by the Year 2010. March 2004
- No. 29 U. Dulleck, N. Foster, R. Stehrer and J. Wörz: Dimensions of Quality Upgrading in CEECs. April 2004
- No. 30 R. Stehrer: Can Trade Explain the Sector Bias of Skill-biased Technical Change? May 2004
- No. 31 M. Landesmann and R. Stehrer: Income Distribution, Technical Change and the Dynamics of International Economic Integration. September 2004
- No. 32 A. Hildebrandt and J. Wörz: Determinants of Industrial Location Patterns in CEECs. November 2004
- No. 33 K. Laski: Macroeconomics versus 'Common Sense'. December 2004
- No. 34 V. Astrov, Sectoral Productivity, Demand, and Terms of Trade: What Drives the Real Appreciation of the East European Currencies? April 2005
- No. 35 R. Stehrer: The Effects of Factor- and Sector-biased Technical Change Revisited. September 2005
- No. 36 N. Foster and R. Stehrer: Modelling GDP in CEECs Using Smooth Transitions. December 2005
- No. 37 J. Francois and J. Wörz: Rags in the High Rent District: The Evolution of Quota Rents in Textiles and Clothing.

 January 2006
- No. 38 C. F. Castejón and J. Wörz: Good or Bad? The Influence of FDI on Output Growth. An industry-level analysis. April 2006
- No. 39 R. Römisch and M. Leibrecht: An Alternative Formulation of the Devereux-Griffith Effective Average Tax Rates for International Investment. May 2006
- No. 40 E. Christie and M. Holzner: What Explains Tax Evasion? An Empirical Assessment based on European Data. June 2006
- No. 41 M. Landesmann and R. Stehrer: Goodwin's Structural Economic Dynamics: Modelling Schumpeterian and Keynesian Insights. October 2006
- No. 42 A. Bhaduri: On the Dynamics of Profit- and Wage-led Growth. March 2007
- No. 43 P. Esposito and R. Stehrer: The Sector Bias of Skill-biased Technical Change and the Rising Skill Premium in Transition Economies. May 2007
- No. 44 R. C. Feenstra: Globalization and Its Impact on Labour. July 2007
- No. 45 K. Laski: Do Increased Private Saving Rates Spur Economic Growth? September 2007
- No. 46 A. Iara: Skill Diffusion by Temporary Migration? Returns to Western European Work Experience in Central and East European Countries. July 2008
- No. 47 N. Foster and R. Stehrer: Sectoral Productivity, Density and Agglomeration in the Wider Europe. September 2008

Neil Foster is Assistant Professor at the University of Vienna, Department of Economics. Robert Stehrer is Research Economist at the Vienna Institute for international Economic Studies (wiiw).

The present paper was prepared as part of the EU FP6 programme project MICRO-DYN, 'The competitiveness of firms, regions and industries in the knowledge-based economy: What room for job-rich growth in Europe?'

Neil Foster and Robert Stehrer

Sectoral Productivity,
Density and
Agglomeration in
the Wider Europe

Contents

Αl	ostract		i
1	Introd	luction	1
2	The r	nodel	4
	2.1	Externalities within regions	4
	2.2	Externalities across regions	6
3	Data		7
4	Resu	Its	9
	4.1	Results for the total economy	9
	4.2	Sectoral results	11
5	Instru	mental Variable results	12
	5.1	Total economy	12
	5.2	Sectoral dimension	13
6	Resu	Its for country subsamples	15
	6.1	Total economy	16
	6.2	Sectoral dimension	16
7	Conc	lusions	19
Αŗ	pendi	x	
	A Matl	nematical appendix	21
		sifications	
D	oforon		22

List of Tables

Table 1	Agglomeration and spillover effects at the aggregate level	. 9
Table 2	Agglomeration and spillover effects at the sectoral level	11
Table 3	Density and spillovers at the aggregate level – IV results	13
Table 4	Agglomeration and spillover effects at the sectoral level – IV results	14
Table 5	Agglomeration and spillover by country groups in the total economy – IV results)	17
Table 6	Density and spillover effects at the sectoral level in the Old and New Member States – IV results	18

Abstract

In this paper we extend the agglomeration model of Ciccone (2002) to the level of industry. We then test this model using panel data for six sectors on regional level data for 27 EU member states. Our results for the aggregate economy confirm the estimates of Ciccone (2002). For our full sample of countries the sectoral level results also indicate significant agglomeration effects, with the exception of agriculture. Considering differences in the extent of agglomeration effects between new and old EU member states, however, leads to the conclusion that agglomeration effects tend to be stronger at both the aggregate and the sectoral level for new member states.

Keywords: agglomeration, employment density, productivity, European regions

JEL classification: R10

1 Introduction

The agglomeration of economic activities in a few locations has been a distinctive feature of the world economy for many years. Despite early attempts at accounting for such agglomeration effects (see for example Marshall, 1920; Mills, 1967; Henderson, 1974) it is only recently that economists have been able to explain such effects (see for example Krugman, 1991; Ciccone, 1992; Krugman and Venables, 1995; Fujita and Thisse, 1996).

Understanding the forces of agglomeration is important since there exists a strong positive correlation between output or productivity growth and geographical agglomeration (see for example Hohenberg and Lees, 1985). In Europe, for instance, output growth during the industrial revolution was accompanied by the formation of industrial clusters in the core of Europe that largely still remain (Baldwin and Martin, 2004). The results of De la Fuente and Vives (1995) suggest that while convergence has taken place within Europe, regions inside countries have either failed to converge or have diverged, results again suggestive of the important complementarities between growth and agglomeration.

Explanations for the observed relationship between agglomeration economies and productivity within countries relate to spatial externalities and to increasing returns at the firm level combined with non-tradabilities or transport costs. Ciccone and Hall (1996) for example develop two models in which the spatial density of economic activity is the source of aggregate increasing returns. The models are based upon either local geographic externalities or upon the variety of local intermediate services. Density is assumed to affect productivity through a number of channels. Firstly, positive transport costs may lead to increasing returns within a geographical area, enhancing productivity in denser regions. Secondly, externalities associated with the physical proximity of production tend to increase productivity in regions with greater density. Such externalities include the possibility of labour market pooling, input sharing and localized technological spillovers. Thirdly, a higher degree of specialization is possible in areas of dense activity.

Few direct tests of agglomeration effects and the relationship between agglomeration and growth exist. Rosenthal and Strange (2003), however, examine the effect of existing employment on both the birth of new firms and employment in new establishments using US data at the zip

code level. They find that agglomeration economies decline rapidly with distance, with the effect of own industry employment in the first mile up to 1000 times larger than the effect 2 to 5 miles away. Further evidence is provided by Dekle and Eaton (1999) who find for a sample of Japanese regions that the density of an industry in a region has a significant impact on productivity in that region, while Rice et al. (2006) find that labour productivity in British regions increases with proximity to economic mass. Most related to the current paper however are the papers by Ciccone and Hall (1996) and Ciccone (2002) to which we now turn.

Ciccone and Hall (1996) use 1988 US county level data to test their model. They construct a measure of state level labour productivity, regressing this variable on a measure of the density of employment in counties within the state and measures of education. The results indicate that doubling the employment density in a county increases labour productivity by 6 per cent on average. Ciccone (2002) conducts a similar exercise for 628 NUTS-3 regions in Germany, Italy, France, Spain and the UK. He regresses a measure of labor productivity, constructed by using data on non-agricultural private value added and salaried employment at the regional level, on the density of employment, education variables, and country and regional dummies. Ciccone finds substantial agglomeration effects for the five European countries, with the results suggesting that a doubling of the employment density increases labour productivity by approximately five per cent, a result similar to that found for the US by Ciccone and Hall (1996).

The current paper closely follows the approach of Ciccone (2002), but adds to this earlier work in a number of ways. Firstly, we extend the country coverage to include information on 255 regions in 26 European countries. Our sample also includes Central and East European countries¹, which allows us to examine whether the agglomeration effects found in the core advanced countries of Europe are also present for a wider range of European countries. Secondly, rather than concentrating on total output, or total output minus agriculture and the public sector, we consider agglomeration effects using sectoral data on six sectors, thus allowing us to examine differences in agglomeration effects across sectors. One should note, however, that from the extension of the model we provide in this paper we cannot hypothesize on the magnitude of the density and spillover effects across various sectors. For this we rely on empirical assessment, though one should also note that there is little empirical evidence on similarities or differences in density and spillover effects across sectors. This paper thus provides some initial insights into this

¹Included are all current EU members with the exception of Malta for data reasons.

issue. Thirdly, while time series data at the regional level on the variables of interest are only available for a short period of time, we do make use of panel data considering agglomeration effects using annual data over the period 1998-2004. Before moving on to discuss the model and the results, it should be mentioned that one recent paper, Brülhart and Mathys (2008), also considers the issue of agglomeration using sectoral panel data. The paper differs from the current one in a number of respects, notably by the use of a dynamic panel model to deal with the issue of endogeneity. In addition, while the paper by Brülhart and Mathys (2008) uses data over a longer period, the current paper has a broader country coverage. The results of Brülhart and Mathys (2008) suggest stronger agglomeration effects at the aggregate level than those found by Ciccone (2002). The authors choose to concentrate on the results from manufacturing and financial services and find in the case of manufacturing that own-sector density lowers productivity, while other-sector density increases productivity. For financial services the reverse is true, with own-sector density raising productivity in the financial sector, but other-sector density reducing it.

Our results for the full sample of countries indicate significant agglomeration effects at the aggregate level. Significant agglomeration effects are also found for five of the six sectors (the exception being agriculture). These results are largely unchanged when we account for endogeneity using Instrumental Variable (IV) techniques. Of perhaps greater interest are the results obtained when we split our sample into 'old' and 'new' Europe. We find agglomeration effects in new Europe to be large and significantly larger than those in old Europe both at the aggregate and sectoral levels. The results suggest that dense regions in new Europe, such as capital cities, may have better prospects for productivity growth when compared with less dense regions. This may reflect the uneven levels of transport and business infrastructure in new Europe as compared to old Europe.

The remainder of the paper is organized as follows. Section 2 describes the theoretical model of Ciccone (2002), and extends it to the case of more than one sector. We then move on to a brief description of our data sources and classifications in Section 3. Section 4 presents results for the aggregate economy and the different sectors, while Section 5 reports results from an instrumental variable approach that deals with a potential endogeneity problem. Section 6 reports results for two country sub-samples, the Western European economies and the Eastern European transition economies. Section 7 concludes.

2 The model

2.1 Externalities within regions

In this section we introduce the basic model used in Ciccone (2002), but extend it to the case of more than one sector. We denote the production function for a particular sector i on an acre of land in a particular region s contained in country c by,

$$q_{isc} = \Omega_{isc} f_i(n_{isc} H_{isc}, k_{isc}, Q_{isc}, A_{sc})$$

where q_{isc} denotes output, n_{isc} the number of workers, H_{isc} the average level of human capital of these workers and k_{isc} the stock of physical capital used on this acre of land in this particular sector. We assume that the average level of human capital differs across sectors. In addition the capital stock is also assumed to be sector-specific which may be explained by differences in sectoral intensities with respect to asset types. This allows us to implicitly model the sectorspecific use of particular factors such as land, etc. or sector-specific capital requirements. The other three parameters are specific to the region and sector with Ω_{isc} denoting an index of total factor productivity, Q_{isc} total output of the region and A_{sc} total acreage of the region. The sectorspecific density of the region is then defined as Q_{isc}/A_{sc} . It is assumed that spatial externalities are driven by the density of production. In our disaggregated framework we assume that it is the sector-specific density of production that matters for agglomeration, which may arise due to sector-specific spillovers or sector-specific labour qualifications, for example. Agglomeration effects in this framework are thus not due to other regional specific effects, such as overall production density or the level of general infrastructure within a region (see Ciccone and Hall, 1996).² Our paper thus considers the importance of localization economies, that is, the benefit a firm gains from locating in close proximity to firms in the same sector, rather than urbanization economies, which are the benefits derived from firms locating in close proximity to other firms from all sectors. This choice is justified by noting that in the existing literature most studies have found that localization effects tend to be most prevalent (see for example the review article of Rosenthal and Strange, 2004).

Under the assumption that the elasticity of output per acre with respect to density is constant,

²These overall effects are captured to an extent by country and regional NUTS-1 dummies in the econometric specification below.

we obtain the following specification,

$$q_{isc} = \Omega_{isc} f_i(n_{isc} H_{isc}, k_{isc}, Q_{isc}, A_{sc}) = \Omega_{isc} \left((n_{isc} H_{isc})^{\beta_i} k_{isc}^{1-\beta_i} \right)^{\alpha_i} \left(\frac{Q_{isc}}{A_{sc}} \right)^{(\gamma_i - 1)/\gamma_i}.$$

where $0 \le \alpha_i \le 1$ captures the returns to capital and labour and $0 \le \beta_i \le 1$ is a distribution parameter. The parameter γ_i denotes sector-specific spatial externalities. Positive spatial externalities are present if and only if $\gamma_i > 1$.

Assuming that labour and capital are distributed equally among the acres in each region, the production function of the region is $Q_{isc} = q_{isc}A_{sc}$ or,

$$Q_{isc} = A_{sc} \Omega_{isc} \big[(\frac{N_{isc}H_{isc}}{A_{sc}})^{\beta_i} \big(\frac{K_{isc}}{A_{sc}}\big)^{1-\beta_i} \big]^{\alpha_i} \big(\frac{Q_{isc}}{A_{sc}}\big)^{(\gamma_i-1)/\gamma_i}.$$

Here N_{isc} is total employment, H_{isc} is the average human capital level of workers employed in sector i, and K_{isc} denotes the total amount of capital in region s and sector i. Solving the latter equation for average labour productivity gives,

$$\frac{Q_{isc}}{N_{isc}} = \Omega_{isc}^{\gamma_i} \left(H_{isc}^{\beta_i} \left(\frac{K_{isc}}{N_{isc}} \right)^{1-\beta_i} \right)^{\alpha_i \gamma_i} \left(\frac{N_{isc}}{A_{sc}} \right)^{\alpha_i \gamma_i - 1}.$$

The appearance of the capital stock K_{isc} in this equation is troublesome since data for this variable are not available. To get around this problem we follow Ciccone (2002) and assume that the rental price of capital r_{ic} is equalized within a country. The capital demand function is then given by,

$$K_{isc} = \frac{\alpha_i(1-\beta_i)}{r_{ic}}Q_{isc}.$$

Substituting this into the expression above and again solving for labour productivity yields,

$$\frac{Q_{isc}}{N_{isc}} = \Gamma_{ic} \Omega_{isc}^{\omega_i} H_{isc} \left(\frac{N_{isc} H_{isc}}{A_{sc}} \right)^{\theta_i}$$

where $\Gamma_{ic} = \left(\frac{\alpha_i(1-\beta_i)}{r_{ic}}\right)^{\frac{(1-\beta_i)\alpha_i\gamma_i}{1-(1-\beta_i)\alpha_i\gamma_i}}$ and $\omega_i = \frac{\gamma_i}{1-(1-\beta_i)\alpha_i\gamma_i}$ are constants. The first is country- and sector-specific as long as the rental price of capital differs across countries for each sector.³ The second constant is sector-specific and the same across countries. The effect of regional density arises through the parameter,

$$\theta_i = \frac{\alpha_i \gamma_i - 1}{1 - \alpha_i \gamma_i (1 - \beta_i)}$$

³Note that we assume the rental price of capital to be country- and sector-specific. This may be justified in the way that the capital stock differs across industries in its intensity with respect to asset types. If the rental prices of these asset types differ (but are equalized across the country) these are sector-specific. If we assume $r_{ic} = r_c$, i.e. rental prices are equalized across sectors within a country, we would reach the same conclusions as Γ_{ic} would still be sector- and country-specific.

to which we will refer as the 'agglomeration effect'. Finally, taking logs yields,

$$\ln \frac{Q_{isc}}{N_{isc}} = \ln \Gamma_{ic} + \omega_i \ln \Omega_{isc} + (\theta_i + 1) \ln H_{isc} + \theta_i \ln (\frac{N_{isc}}{A_{sc}}). \tag{2.1}$$

Thus sectoral labour productivity is a function of sectoral total factor productivity, the average level of sectoral human capital and the sector-specific employment density in the particular region. In the econometric implementation we follow Ciccone (2002) and use the share of high- and low-skilled workers in the particular sectors as a measure of the level of human capital.⁵ The constants in equation (2.1) are accounted for by including country and NUTS-1 regional dummies. We also include a set of time dummies in our equation to account for time-specific effects and include sector dummies when estimating the model at the sectoral level. Our econometric specification which is estimated for each particular sector is thus,

$$\ln \frac{Q_{isc}}{N_{isc}} = D_c \beta_c + D_t \beta_t + D_r \beta_r + \sum_{e=H,L} \delta_{ei} h_{eisc} + \theta_i \ln(\frac{N_{isc}}{A_{sc}}) + u_{isc}.$$
 (2.2)

where $D_j(j=c,t,r)$ denote dummies for countries, time periods and NUTS-1 regions, and β_j denote the coefficients to be estimated. The share of workers with tertiary (e=H) and primary (e=L) educational attainment levels are denoted by h_{eisc} with the coefficients denoted by δ_{ei} . θ_i is the coefficient capturing the agglomeration effect.

2.2 Externalities across regions

The model thus far only captures sectoral level externalities within regions. As Ciccone (2002) noted however, there are goods reasons to believe that spillovers may extend beyond regions. Indeed, research has shown that the process of convergence in European regions displays spatial dependence. Ertur and Baumont (2006) for example show that spatial dependence is important in the estimation of β convergence among 138 European regions. To account for cross regional spillovers Ciccone (2002) allowed total factor productivity in a region to be dependent upon the density of production in neighbouring regions. We follow this approach assuming that total factor productivity Ω_{isc} is a function of production density in 'neighbouring' regions. Formally, this means that,

$$\Omega_{isc} = \Phi_{isc} \left(\frac{Q_{isc}^n}{A_{sc}^n} \right)^{\mu_i}$$
.

⁴Ciccone (2002) discusses in detail the conditions under which $\theta_i > 0$.

⁵From our data we can only distinguish three types of educational attainment levels.

where Φ_{isc} denotes total factor productivity which is assumed to be exogenous. Output in the neighbouring regions is denoted by $Q_{isc}^n = \sum_{r,d} d_{rd,sc} Q_{ird}$ and the total acreage of neighbouring regions is $A_{sc}^n = \sum_{r,d} d_{rd,sc} A_{rd}$. Here $d_{rd,sc}$ denotes a distance measure of region s in country c to all other regions. Inserting $\ln \Omega_{isc} = \ln \Phi_{isc} + \mu_i \ln \left(\frac{Q_{isc}^n}{A_{sc}^n}\right)$ into equation (2.1) above yields

$$\ln \frac{Q_{isc}}{N_{isc}} = \ln \Gamma_{ic} + \omega_i \ln \Phi_{isc} + \omega_i \mu_i \ln \left(\frac{Q_{isc}^n}{A_{nc}^n}\right) + (\theta_i + 1) \ln H_{isc} + \theta_i \ln \left(\frac{N_{isc}}{A_{sc}}\right).$$

This results in an econometric specification similar to equation (2.2) above,

$$\ln \frac{Q_{isc}}{N_{isc}} = D_c \beta_c + D_t \beta_t + D_r \beta_r + \sum_{e=H,L} \delta_{ei} h_{eisc} + \theta_i \ln(\frac{N_{isc}}{A_{sc}}) + \varphi_i \ln(\frac{Q_{isc}^n}{A_{sc}^n}) + u_{isc}$$
 (2.3)

which differs only by the appearance of the effect of production density in 'neighbouring' regions with $\varphi_i = \omega_i \mu_i$ being the coefficient to be estimated.

Rather than only considering spillovers among regions sharing a common border, as is done in Ciccone (2002), we also consider alternative weighting matrices. Apart from a common border matrix, we apply an exponential decay function with different decay parameters ϕ ,

$$d_{ij} = \begin{cases} exp^{-\phi \operatorname{dist}_{ij}} & \text{for } i \neq j \\ 0 & \text{for } i = j \end{cases}$$

Here dist_{ij} denotes the great circle distance between regions i and j. The distance between two regions is measured from the centre of each region in kilometres and is divided by 1000 to allow us to calculate the exponential. We report results for decay parameters $\phi = 10, 5, 2.5, 1, 0.25$. The larger the decay parameter, the less weight is given to regions further away. The product of each of these weighting matrices and the density of the production matrix provides us with a variable measuring cross-regional spillovers.⁶

3 Data

The data we are using are taken from the Eurostat Regio database. We use value added and (total) employment data from Regional National Accounts over the period 1998-2003. For some countries and years we had to impute missing values by linear interpolation to give a balanced

⁶In the specifications reported below we set $d_{ii} = 0$. We also experimented, however, by including internal distances (measured by replacing zero with an internal distance measure given by $\operatorname{dist}_{ii} = \sqrt{A_i/\pi}$). These results are not reported for brevity, but are available upon request.

panel. The value added data, which are in nominal terms, have been deflated (base year 2003) using country- and sector-specific deflators since no regional deflators are available. The data were also converted to a common currency using 2003 PPPs. The National Accounts employment data are collected according to the 'principle of working condition', that is, people are counted in the region where they are working and not where they are living. Given data on output by region and sector this allows one to calculate labour productivity levels. We also include the shares of high- and low-skilled workers (defined by educational attainment levels) by sector and region in the regressions. These data are taken from the Labour Force Survey (LFS), which is (consistently) available for the period 1998-2005. The LFS data provide information on educational attainment levels according to the International Standard Classification of Education (ISCED) categories and allow for three types of worker (high-, medium- and low-skilled). As mentioned previously, we include the share of high- and low-skilled as measures of the level of human capital.

In the analysis we are restricted to six sectors according to the NACE classification. These are: Agriculture, Forestry and Fishing (AtB), Mining, Manufacturing and Energy (CtE), Construction (F), Wholesale and Retail Trade, Transport (GtI), Business Services (JtK) and Public Services (LtQ) (see appendix Table 7 for a more detailed description). Finally, the number of regions considered according to the NUTS-2 classification is 255, with data from 26 European countries included.⁸ Data on geographical area are taken from Eurostat. Compared with the study of Ciccone (2002) the current paper thus has a broader coverage of countries within Europe as well as more disaggregated data, considering the importance of agglomeration across six broad economic sectors.⁹ On the other hand, the data in the present paper are less detailed with respect to the regional classification as Ciccone (2002) applies data at the NUTS-3 level whereas we use data at the NUTS-2 level.

⁷LFS data are collected according to the 'principle of living condition', that is, people are counted in the region where they are living. This introduces a slight inconsistency in our dataset which cannot be resolved. Using employment levels from the LFS data (instead of employment levels from Regional National Accounts data), however, yields similar results.

⁸The countries considered in our analysis are: Austria, Belgium, Bulgaria, Cyprus, Czech Republic, Denmark, Estonia, Finland, France, Germany, Greece, Spain, Hungary, Ireland, Italy, Latvia, Lithuania, Luxembourg, Netherlands, Poland, Portugal, Romania, Slovakia, Slovenia, Sweden, and the UK.

⁹Ciccone (2002) did not consider the non-agricultural and private economy in his aggregate data on value-added.

4 Results

In this section we present our basic results. We do this using aggregate data first and then for the sectoral data.

4.1 Results for the total economy

The model given by equations (2.2) and (2.3) is estimated, first, for the total economy and for the total economy minus agriculture, forestry and fishing (AtB) and public services (LtQ), which we refer to as the 'total market economy'. This allows us to compare the results of Ciccone (2002) using NUTS-3 level data with our results using data at the NUTS-2 level. The results are reported in Table 1. The top half of this table reports the results for the total economy, while the lower half reports the results for the total market economy. The column headings refer

	Intra	Common					
Variable	Spillovers	Border	$\phi = 10.00$	$\phi = 5.00$	$\phi = 2.50$	$\phi = 1.00$	$\phi = 0.25$
				Total econo			
Density	0.050 ***	0.049 ***	0.048 ***	0.048 ***	0.049 ***	0.049 ***	0.051 ***
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
High skill	0.495	0.506	0.554	0.526	0.504	0.508	0.507
	(0.239)	(0.230)	(0.191)	(0.219)	(0.240)	(0.237)	(0.235)
Low skill	-0.935 ***	-0.937 ***	-0.964 ***	-0.947 ***	-0.937 ***	-0.937 ***	-0.932 ***
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Spillovers		0.020 **	0.067 ***	0.045 **	0.023	0.115	0.866
		(0.027)	(0.000)	(0.045)	(0.548)	(0.340)	(0.101)
\overline{F}	437.361	402.359	425.717	432.487	433.109	434.310	437.091
R^2	0.933	0.933	0.934	0.933	0.933	0.933	0.933
Obs.	1530	1506	1530	1530	1530	1530	1530
			То	tal market ec	onomy		
Density	0.048 ***	0.048 ***	0.047 ***	0.047 ***	0.049 ***	0.048 ***	0.048 ***
Delisity	(0.000)	(0.000)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)
High skill	0.439	0.447	0.459	0.444	0.433	0.437	0.441
mgn skin	(0.463)	(0.456)	(0.445)	(0.461)	(0.470)	(0.467)	(0.441)
Low skill	-0.312	-0.292	-0.340	-0.318	-0.302	-0.307	-0.316
LOW SKIII							
G :11	(0.202)	(0.239)	(0.164)	(0.190)	(0.210)	(0.203)	(0.192)
Spillovers		0.011	0.037 **	0.012	-0.035	-0.057	0.303
		(0.271)	(0.026)	(0.619)	(0.396)	(0.654)	(0.587)
F_{\perp}	489.042	485.633	488.118	484.740	482.733	482.330	486.680
R^2	0.913	0.913	0.913	0.913	0.913	0.913	0.913
Obs.	1530	1506	1530	1530	1530	1530	1530

All regressions include time, country and NUTS1 dummies; p-values in brackets are based on heteroscedasticity consistent standard errors; ***, ** denote significance at the 1, 5, and 10 % level.

Table 1: Agglomeration and spillover effects at the aggregate level

to the different weighting schemes considered when accounting for cross-regional spillovers. The results on our main variable of interest, density, are largely consistent across specifications and

for the two aggregates. In particular, we find a coefficient around 0.05 for both the total and the total market economy. As with Ciccone (2002) we find the results on the density variable to be robust to the different weighting schemes for regional spillovers. Despite the problems in comparing results at different spatial levels, the coefficients on density at the NUTS-2 level are remarkably similar to those found by Ciccone (2002) at the NUTS-3 level and suggest that a doubling of employment density increases labour productivity by about 5 per cent.¹⁰

In terms of the additional variables in the model, the coefficient on high-skilled education is positive, though not significant, while that on low-skilled education is negative and significant in the case of the total economy at least. The lack of significance of this variable is consistent with findings in the growth literature where human capital and growth have found to be only weakly, or even negatively, related (see for example Benhabib and Spiegel, 1994; Pritchett, 2001). Finally, we consider the coefficients on the spillover variables. For the common border variable we find a coefficient of 0.02 for the total economy, which is significant, and 0.01 for the total market economy. These coefficients are smaller than those found by Ciccone (2002) who finds a coefficient of 3.3 per cent. This may reflect the Modifiable Areal Unit Problem with the distances in this paper being much larger than those in Ciccone (2002), which may lower the importance of spillovers. Considering the spillover variable for different values of ϕ , we tend to find that the coefficient is positive, significant and relatively large for high values of ϕ becoming smaller (and sometimes negative in the case of the market economy) and insignificant for smaller values of ϕ . These results are consistent with existing results indicating that spillovers decline rapidly with distance.¹¹

¹⁰In particular, the modifiable areal unit problem suggests that the choice of spatial scale is likely to affect the results. Moreover, we may expect larger spillovers at the NUTS-3 level due to the average distance between regions at the NUTS-3 level being much larger. As discussed above, existing literature suggests that spillovers decline rapidly with distance.

¹¹As mentioned above, in some specifications we also include internal distances in the spillover variable to account for the fact that internal distances at the NUTS-2 level can be quite large. When doing this the results for density tend to be consistent with those reported above, while the coefficients on the spillover variables tend to increase, reinforcing the short-distance nature of spillovers. Results are available on request.

4.2 Sectoral Results

Table 2 reports results for density and spillover effects in the different sectors.¹² The coefficients on density, the share of high- and low-skilled workers and, where relevant, cross-regional spillovers are estimated separately for each sector, thus allowing the coefficients on the dummies to vary across sectors. The coefficients on the density variable in Table 2 (Column 1) indicate that there

		Intra	Common					
Variable		Spillovers	Border	$\theta = 10.00$	$\theta = 5.00$	$\theta = 2.50$	$\theta = 1.00$	$\theta = 0.25$
Density	Agriculture	-0.032***	-0.036***	-0.040***	-0.048***	-0.055***	-0.053***	-0.031***
v	Ü	(0.001)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.001)
	Manufacturing	0.042***	0.042***	0.040***	0.040***	0.040***	0.039***	0.044***
		(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	Construction	0.057***	0.057***	0.056***	0.057***	0.060***	0.060***	0.056***
		(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	Trade	0.057***	0.056***	0.056***	0.056***	0.057***	0.056***	0.059***
		(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	Business services	0.057***	0.057***	0.056***	0.056***	0.057***	0.057***	0.058***
		(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	Public services	0.055***	0.054***	0.053***	0.053***	0.054***	0.054***	0.057***
		(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Spillover	Agriculture		0.073***	0.120***	0.174***	0.296***	0.927***	3.010***
			(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	Manufacturing		0.006	0.056***	0.049**	0.045	0.245**	1.591***
			(0.487)	(0.000)	(0.020)	(0.223)	(0.034)	(0.001)
	Construction		0.008	0.036**	-0.001	-0.069*	-0.292***	-1.093**
			(0.442)	(0.022)	(0.955)	(0.054)	(0.009)	(0.026)
	Trade		0.018**	0.049***	0.031	0.009	0.103	1.115**
			(0.035)	(0.001)	(0.184)	(0.826)	(0.435)	(0.037)
	Business services		0.010	0.056***	0.035	0.005	0.037	0.512
			(0.140)	(0.000)	(0.111)	(0.896)	(0.736)	(0.253)
	Public services		0.037***	0.093***	0.058**	0.029	0.113	1.041**
			(0.001)	(0.000)	(0.018)	(0.464)	(0.338)	(0.036)

All regressions include time, country and NUTS1 dummies; p-values in brackets are based on heteroscedasticity consistent standard errors; ***, ** denote significance at the 1, 5, and 10 % level.

Table 2: Agglomeration and spillover effects at the sectoral level

are significant agglomeration effects in five of the six sectors. Only in the case of agriculture do we find a significantly negative coefficient. For this industry therefore we find evidence of a congestion effect, which may reflect the smaller average land holdings of farmers in denser regions that may limit the ability to exploit economies of scale. Where positive the size of the coefficients are also in line with those reported at the aggregate level both in this paper and in previous studies, suggesting that a doubling of employment density in a particular sector is associated with an increase in labour productivity of around 5.5 per cent. The size of the coefficient also tends to be fairly similar across industries, though somewhat lower for manufacturing.

Introducing spillover effects (Columns 2-7) has no discernible impact on the size, sign and sig-

¹²The full set of results is available upon request.

nificance of the coefficients on density. The coefficients on the spillover variables themselves tend to differ across sectors. While we find positive and significant spillover effects for all specifications in agriculture, this is not the case for other industries. The results do suggest, however, that the spillover variable is positive and significant for all sectors for high values of θ , again suggesting that spillovers are important, but that their importance declines quickly with distance.

5 Instrumental Variable Results

Ciccone (2002) argues that there are two competing explanations for the agglomeration effects he finds. The first of these suggests that productivity is high because of agglomeration effects, while the second would argue that agglomeration is an outcome of high productivity and not a cause. In the absence of information on all variables causing productivity, Ciccone proceeds to employ an instrumental variables approach to deal with this problem using land area as an instrument for employment density. Ciccone (2002) shows that land area and employment density are negatively correlated after controlling for country fixed effects, a correlation which, Ciccone argues, relates to history. In particular, he argues that the size of regions was determined largely for administrative reasons, with the equalization of the population a natural criterion. Given the historical nature of the correlation between area and employment density, area can be used as an instrument for employment density as long as the original cause of cross-region differences in area affects productivity today largely through agglomeration. Ciccone (2002) finds that using area as an instrumental variable for employment density slightly lowers the estimate of agglomeration effects, but the coefficient is still found to be positive and highly significant. We follow the approach of Ciccone (2002) and use area as an instrument for employment density. 13 We thus proceed to re-estimate the above models using area as an instrument for employment density.

5.1 Total economy

Table 3 presents the results for the total economy and the total market economy. Consistent with the results of Ciccone (2002) we find in Table 3 in most cases a lower coefficient on the density variable when using the instrumental variables approach as compared with the OLS

¹³Regressing employment density on area, country and time dummies at both the aggregate and sectoral levels results in a significant negative association between density and area. These results are available upon request.

results reported in Table 1. The coefficients tend to remain significant for the total economy, but are often insignificant when the total market economy is considered. The coefficients on the education variables are also generally consistent with those from the OLS estimation, with negative coefficients found for the share of low-educated workers and positive, though usually insignificant coefficients found for the share of high-educated workers. Finally, in terms of the spillover variable, we find results again consistent with those in Table 1 with a positive and significant coefficient found for the common border weighting as well as for high values of θ , but insignificant coefficients found for lower values of θ .¹⁴

	Intra	Common					
Variable	Spillovers	Border	$\theta = 10.00$	$\theta = 5.00$	$\theta = 2.50$	$\theta = 1.00$	$\theta = 0.25$
				Total econo	my		
Density	0.041 ***	0.027 *	0.009	0.015	0.030	0.033 *	0.045 ***
	(0.002)	(0.085)	(0.607)	(0.510)	(0.199)	(0.058)	(0.001)
High skill	0.577	0.703 *	0.914 **	0.838 *	0.685	0.651	0.559
	(0.135)	(0.066)	(0.031)	(0.064)	(0.129)	(0.123)	(0.141)
Low skill	-0.936 ***	-0.943 ***	-0.972 ***	-0.955 ***	-0.941 ***	-0.938 ***	-0.933 ***
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Spillovers		0.025 **	0.076 ***	0.064 **	0.047	0.163	0.797
		(0.016)	(0.000)	(0.015)	(0.353)	(0.211)	(0.186)
\overline{F}	429.256	389.603	411.794	419.541	423.243	425.428	429.621
R^2	0.933	0.932	0.931	0.931	0.932	0.932	0.933
Obs.	1530	1506	1530	1530	1530	1530	1530
			To	tal market ec	conomy		
Density	0.015	0.003	-0.004	0.001	0.014	0.013	0.015
v	(0.310)	(0.858)	(0.846)	(0.954)	(0.566)	(0.497)	(0.313)
High skill	0.721	0.831	0.901	0.844	0.734	0.740	0.723
	(0.227)	(0.163)	(0.154)	(0.192)	(0.248)	(0.233)	(0.217)
Low skill	-0.286	-0.264	-0.306	-0.293	-0.287	-0.289	-0.284
	(0.260)	(0.313)	(0.244)	(0.259)	(0.259)	(0.255)	(0.253)
Spillovers	, ,	0.021 *	0.046 ***	0.034	0.007	0.053	-0.089
		(0.062)	(0.005)	(0.213)	(0.899)	(0.692)	(0.890)
\overline{F}	436.620	427.187	414.179	418.425	430.549	430.591	431.426
R^2	0.910	0.908	0.907	0.908	0.910	0.910	0.910
Obs.	1530	1506	1530	1530	1530	1530	1530

All regressions include time, country and NUTS1 dummies; p-values in brackets are based on heteroscedasticity consistent standard errors; ***, ** denote significance at the 1, 5, and 10 % level.

Table 3: Density and spillovers at the aggregate level - IV results

5.2 Sectoral dimension

Table 4 presents the results from the IV estimation for the individual sectors. The results, when excluding spillovers (Column 1), tend to be consistent with the aggregate results and

¹⁴For reasons of brevity we choose not to report the results when including internal distances in the spillover specification. These results are, however, available upon request.

		$_{ m Intra}$	Common					
Variable		Spillovers	Border	$\theta = 10.00$	$\theta = 5.00$	$\theta = 2.50$	$\theta = 1.00$	$\theta = 0.25$
Density	Agriculture	0.060***	0.025	0.032	0.019	0.011	0.017	0.065***
		(0.008)	(0.263)	(0.201)	(0.590)	(0.824)	(0.588)	(0.007)
	Manufacturing	0.045***	0.042**	0.013	0.014	0.028	0.030	0.055***
		(0.003)	(0.020)	(0.541)	(0.593)	(0.347)	(0.158)	(0.001)
	Construction	0.039***	0.032**	0.021	0.034*	0.053***	0.051***	0.034***
		(0.002)	(0.033)	(0.206)	(0.077)	(0.008)	(0.001)	(0.007)
	Trade	0.039***	0.025*	0.017	0.019	0.028	0.030*	0.043***
		(0.002)	(0.089)	(0.306)	(0.344)	(0.198)	(0.067)	(0.001)
	Business services	0.039***	0.031**	0.014	0.018	0.032	0.035**	0.041***
		(0.002)	(0.027)	(0.408)	(0.369)	(0.122)	(0.028)	(0.002)
	Public services	0.041***	0.018	0.002	0.012	0.028	0.033**	0.046***
		(0.002)	(0.251)	(0.931)	(0.590)	(0.186)	(0.044)	(0.001)
Spillover	Agriculture		0.064***	0.082***	0.099**	0.160	0.567***	3.140***
			(0.000)	(0.006)	(0.044)	(0.140)	(0.003)	(0.000)
	Manufacturing		0.006	0.067***	0.067**	0.064	0.281*	1.692***
			(0.587)	(0.000)	(0.027)	(0.319)	(0.058)	(0.002)
	Construction		0.015	0.048***	0.013	-0.061	-0.260**	-1.213**
			(0.256)	(0.007)	(0.625)	(0.202)	(0.036)	(0.019)
	Trade		0.026***	0.056***	0.049*	0.046	0.191	0.896
			(0.010)	(0.000)	(0.055)	(0.373)	(0.160)	(0.159)
	Business services		0.014*	0.060***	0.050**	0.032	0.097	0.324
			(0.055)	(0.000)	(0.034)	(0.480)	(0.399)	(0.550)
	Public services		0.041***	0.100***	0.076***	0.055	0.166	0.908
			(0.001)	(0.000)	(0.005)	(0.236)	(0.174)	(0.125)

All regressions include time, country and NUTS1 dummies; p-values in brackets are based on heteroscedasticity consistent standard errors; ***, ** denote significance at the 1, 5, and 10 % level.

Table 4: Agglomeration and spillover effects at the sectoral level - IV results

existing literature in that the coefficient on the density variable is smaller in most cases, though the coefficients remain positive and significant. The main difference between the OLS and IV results at the sectoral level is that the coefficient on agriculture, which in the OLS case was negative and significant, becomes positive and significant in the IV case. This suggests that the endogeneity problem is rather strong for the agricultural sector. When controlling for the potential endogeneity that higher productivity regions (generally urban areas with a low share of agriculture and relatively low productivity in agriculture as economies of scale cannot be exploited) attract more workers (i.e. a migration from agricultural regions to metropolitan areas) we also find significant density effects for the agricultural sector. As large units of land are necessary to exploit the economies of scale in agriculture, and land area is fixed, migration allows the exploitation of these economies of scale.

Turning to the results when spillovers are allowed we again find that the coefficients on the density variables are generally smaller in size than the corresponding OLS coefficients. Moreover, while the coefficients are always significant when spillovers are excluded and for a value of θ equal to 0.25, for other specifications of the spillover variable the density coefficient is often

insignificant, though always positive. The coefficients on the spillover variables themselves are largely consistent with the OLS results. In particular, we tend to find positive and significant coefficients across sectors for the common border weighting and for high values of θ , but largely insignificant coefficients for low values of θ .

6 Results for Country Subsamples

A relevant question, given the above results and given our comparison with the existing studies of Ciccone and Hall (1996) and Ciccone (2002), is whether agglomeration effects are similar across countries or groups of countries and, in particular, whether agglomeration effects found in advanced countries are also present in the new EU member states. 15 To test for such differences we estimated the model above on two sub-samples: 'old Europe' (EU-15) and 'new Europe' (EU-12). 16 Though it may not be clear why one should expect differences in the density effects between old and new member states, there are reasons for density to have larger impacts on productivity levels in the new member states. For example, economic activity seems to be more concentrated in the regions bordering Western European countries and to be relatively more concentrated in urban areas, particularly capital cities (when compared with Western European regions). The bordering regions and the capital cities have profited from trade integration and foreign direct investment flows to a much much larger extent than other regions in this phase of transition. Similarly, productivity levels in these particular regions relative to overall productivity levels appear to be relatively high when compared with the ratios in old member states. This may be due to some regions in the new member states still having large shares of agriculture or a less favourable industrial structure. Also, productivity increases are caused by large inflows of foreign capital with positive effects on productivity through technology transfer and a larger capital stock. Foreign direct investment can lead to significant clustering activities (e.g. the automobile cluster in the Slovak Republic) and thus lead to significant spillover effects (see for example Campos and Kinoshita, 2003; Kim et al., 2003; Pelegrín and Bolancé, 2008). Finally, note that employment rates in these particular regions are in most cases higher than in agricultural

¹⁵The group of new member states (i.e. countries that became EU members in 2004 or thereafter) consists of Bulgaria, Cyprus, the Czech Republic, Estonia, Hungary, Latvia, Lithuania, Poland, Romania, the Slovak Republic and Slovenia.

¹⁶It should be noted that we still allow for spillovers across regions from the total sample.

regions for example, again strengthening the effect of employment density on productivity. There is, however, scarce empirical literature examining whether density effects differ across different types of regions, though Anastassova (2006) reports differences in density effects for metropolitan versus non-metropolitan regions with the effects being about two times larger for metropolitan (urban) regions. These leads us to expect density effects to be larger in the regions of the new member states.¹⁷

6.1 Total economy

Table 5 reports for both groups of countries the results for the total economy and total market economy respectively. For brevity we only report the IV results.¹⁸ The results from the two country sub-samples suggest large differences in the extent of agglomeration. For old Europe we tend to find coefficients on the density variable that are either small, positive and insignificant or small, negative and significant. For new Europe, however, we find coefficients that are large, positive and significant. The size of the coefficients for new Europe tends to be around three times as large as those reported in Table 1. The results at the aggregate level therefore suggest that, while agglomeration effects in old Europe tend to be small at best, in new Europe they are strong and significant. The results on the new member states thus support our expectations though the strength of the effect is surprisingly high.¹⁹

6.2 Sectoral dimension

The sectoral results for density and spillover effects when splitting the sample into the different country groupings are reported in Table 6. We again choose to report the IV results only. The results at the sectoral level are largely consistent with those from the aggregate level. In particular, the coefficients on density are generally small and insignificant for old Europe, but large and positive for new Europe. Once again the size of the coefficients in new Europe tends to be around three times the size of the coefficients reported in Table 2 and by Ciccone (2002). For

¹⁷In this respect, most countries among the new member states are rather small and in some cases the country as a whole (i.e. including the capital city) is counted as the NUTS 2 region.

¹⁸The OLS results are available upon request. The coefficients from the OLS regressions are generally larger in size and more often significant for both groups of countries.

¹⁹These results also suggest that it may be interesting to distinguish the regions according to various typologies to test for differences in density effects by type of regions. We leave this possibility to future research.

Variable	Intra Spillovers	Common Border	$\theta = 10.00$	$\theta = 5.00$	$\theta = 2.50$	$\theta = 1.00$	$\theta = 0.25$
D : t	0.000	0.004		•	lember States		0.000
Density	0.009	-0.004	-0.011	-0.004	0.003	0.006	0.006
II:mb al-ill	(0.372) 0.846 ***	(0.767) 0.959 ***	(0.470) $1.075 ***$	(0.813) 0.982 ***	(0.886) $0.907 ***$	(0.671) 0.879 ***	(0.581) 0.855 ***
High skill					(0.000)		
Low skill	(0.000) -0.553 ***	(0.000) -0.543 ***	(0.000) -0.546 ***	(0.000) -0.547 ***	(0.000) -0.550 ***	(0.000) -0.552 ***	(0.000) -0.551 ***
LOW SKIII	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Spillovers	(0.000)	0.021 **	0.043 ***	0.027	0.022	0.061	-0.407
opinovers		(0.021)	(0.003)	(0.166)	(0.557)	(0.512)	(0.279)
\overline{F}	660.759	526.658	703.777	677.465	663.438	661.861	645.485
R^2	0.807	0.782	0.795	0.799	0.804	0.806	0.806
Obs.	1206	1188	1206	1206	1206	1206	1206
				•	Iember States		
Density	0.202 ***	0.190 ***	0.191 ***	0.204 ***	0.204 ***	0.200 ***	0.204 ***
	(0.000)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.000)
High skill	-0.659	-0.501	-0.517	-0.650	-0.670	-0.645	-0.664
	(0.496)	(0.614)	(0.594)	(0.504)	(0.494)	(0.514)	(0.500)
Low skill	-1.323 ***	-1.326 ***	-1.446 ***	-1.336 ***	-1.291 ***	-1.295 ***	-1.309 ***
~	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Spillovers		0.036	0.207 ***	0.217 ***	0.256 **	0.554 **	1.909
		(0.328)	(0.000)	(0.001)	(0.013)	(0.039)	(0.113)
F_{α}	95.520	96.096	106.478	100.196	96.390	95.480	93.924
R^2	0.876	0.874	0.884	0.882	0.880	0.879	0.878
Obs.	324	318	324	324	324	324	324
			Total market	economy: Ol	ld Member St	ates	
Density	-0.021 *	-0.030 *	-0.032 **	-0.022	-0.013	-0.016	-0.030 *
v	(0.078)	(0.066)	(0.043)	(0.206)	(0.493)	(0.248)	(0.051)
High skill	1.224 ***	1.318 ***	1.342 ***	1.235 ***	1.149 ***	1.183 ***	1.270 ***
_	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Low skill	-0.168 *	-0.134	-0.168 *	-0.167 *	-0.169 *	-0.165 *	-0.140
	(0.081)	(0.167)	(0.087)	(0.082)	(0.075)	(0.084)	(0.164)
Spillovers		0.012	0.023	0.002	-0.029	-0.084	-1.380 ***
		(0.220)	(0.120)	(0.911)	(0.460)	(0.384)	(0.002)
\overline{F}	320.578	252.593	292.941	314.472	329.894	324.525	311.290
R^2	0.794	0.777	0.783	0.793	0.801	0.798	0.787
Obs.	1206	1188	1206	1206	1206	1206	1206
			T-4-1l4		М Си		
Density	0.201 ***	0.193 ***	0.201 ***	0.205 ***	ew Member St 0.201 ***	0.198 ***	0.202 ***
Density	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
High skill	-0.903	-0.803	-0.833	-0.891	-0.902	-0.893	-0.910
mgn skm	(0.322)	(0.400)	(0.362)	(0.326)	(0.325)	(0.335)	(0.324)
Low skill	-0.741 *	-0.749 *	-0.796 *	-0.643	-0.612	-0.669	-0.700
250 5000	(0.090)	(0.082)	(0.059)	(0.118)	(0.137)	(0.111)	(0.101)
Spillovers	(0.000)	0.043	0.192 ***	0.237 ***	0.300 ***	0.654 **	2.319 **
F		(0.252)	(0.001)	(0.000)	(0.003)	(0.012)	(0.047)
\overline{F}	62.683	57.098	67.840	64.975	62.164	61.483	60.896
R^2	0.810	0.804	0.818	0.819	0.816	0.815	0.812
Obs.	324	318	324	324	324	324	324

All regressions include time, country and NUTS1 dummies; p-values in brackets are based on heteroscedasticity consistent standard errors; ***,**, * denote significance at the 1, 5, and 10 % level.

Table 5: Agglomeration and spillover by country groups in the total economy (IV results)

Variable	Industry	Intra Spillovers	Common Border	$\theta = 10.00$	$\theta = 5.00$	$\theta = 2.50$	$\theta = 1.00$	$\theta = 0.25$
	J. Committee of the com	1						
				d Member S				
Density	Agriculture	0.011	0.002	0.009	0.005	-0.009	-0.012	0.017
	3.6	(0.388)	(0.916)	(0.613)	(0.841)	(0.829)	(0.590)	(0.162)
	Manufacturing	0.010	0.012	-0.004	0.009	0.013	0.008	0.007
	C	(0.369)	(0.404)	(0.831)	(0.647)	(0.622)	(0.645)	(0.545)
	Construction	0.008	-0.002	-0.007	0.005	0.014	0.013	0.005
	Trade	(0.372)	(0.875)	(0.636)	(0.754)	(0.394)	(0.295)	(0.658)
	rrade	0.008 (0.370)	-0.008 (0.583)	-0.011 (0.440)	-0.008 (0.654)	-0.002 (0.915)	0.004 (0.783)	0.007 (0.518)
	Business services	0.008	-0.004	-0.007	-0.003	0.003	0.006	0.004
	Dusiliess services	(0.363)	(0.770)	(0.596)	(0.857)	(0.863)	(0.608)	(0.693)
	Public services	0.009	-0.010	-0.016	-0.007	0.002	0.005	0.008
	1 ublic services	(0.374)	(0.499)	(0.318)	(0.674)	(0.919)	(0.671)	(0.486)
G :11	A : 1,	(0.314)			,		, ,	
Spillover	Agriculture		0.017	0.007	0.015	0.066	0.334**	1.794***
	Manufacturing		(0.140) -0.004	$(0.718) \\ 0.024*$	$(0.687) \\ 0.000$	(0.473) -0.012	(0.046) 0.032	(0.000) -0.380
	Manufacturing		(0.640)	(0.024)	(0.987)	(0.848)	(0.805)	(0.307)
	Construction		0.019	0.036**	0.008	-0.020	-0.089	-0.729**
	Collisti uction		(0.103)	(0.021)	(0.688)	(0.568)	(0.324)	(0.040)
	Trade		0.027***	0.043***	0.035*	0.039	0.098	-0.274
	Trade		(0.004)	(0.002)	(0.079)	(0.336)	(0.328)	(0.487)
	Business services		0.019**	0.033***	0.023	0.020	0.046	-0.577*
	Dusiness services		(0.014)	(0.008)	(0.171)	(0.546)	(0.574)	(0.086)
	Public services		0.030***	0.054***	0.036*	0.027	0.069	-0.169
	1 45110 501 11005		(0.005)	(0.000)	(0.070)	(0.440)	(0.430)	(0.658)
				w Member S				
Density	Agriculture	1.097*	1.025	1.153	1.290	1.587	1.904	1.543
		(0.065)	(0.101)	(0.119)	(0.146)	(0.203)	(0.241)	(0.134)
	Manufacturing	0.213***	0.222***	0.193***	0.209***	0.214***	0.211***	0.216***
		(0.002)	(0.004)	(0.006)	(0.003)	(0.003)	(0.003)	(0.003)
	Construction	0.181***	0.173***	0.175***	0.184***	0.183***	0.179***	0.182***
	m 1	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	Trade	0.177***	0.174***	0.178***	0.185***	0.180***	0.175***	0.178***
	ъ.	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	Business services	0.177***	0.170***	0.175***	0.180***	0.179***	0.176***	0.179***
	Public services	(0.000) $0.178***$	(0.000) $0.162***$	(0.000) 0.180***	(0.000) $0.185***$	(0.000) $0.181***$	(0.000)	(0.000) 0.181***
	Public services	(0.000)	(0.000)	(0.000)		(0.000)	0.177***	(0.000)
	A . 1.	(0.000)			(0.000)		(0.000)	
Spillover	Agriculture		0.103	-0.180	-0.841	-2.747	-8.494	-17.097
	3.6		(0.638)	(0.727)	(0.439)	(0.316)	(0.280)	(0.184)
	Manufacturing		-0.030	0.161**	0.225***	0.268**	0.612**	2.522*
	C		(0.513)	(0.019)	(0.003)	(0.015)	(0.032)	(0.050)
	Construction		0.036	0.158***	0.209***	0.288***	0.623**	2.021*
	T 1-		(0.210)	(0.000)	(0.000)	(0.003)	(0.015)	(0.069)
	Trade		0.008	0.166***	0.228***	0.280**	0.580**	1.714
	Business services		(0.787)	(0.006)	(0.003)	(0.012)	(0.038)	(0.167) $1.660*$
	Dusiness services		(0.202)	0.132***	0.148***	0.193**	0.445**	
	Dublia asseriass		(0.302) $0.103***$	(0.000) $0.229***$	(0.002) $0.207***$	(0.015) $0.267***$	(0.037) $0.648***$	(0.081) $2.426**$
	Public services		(0.005)	(0.000)	(0.000)	(0.003)	(0.008)	(0.028)
			(0.000)	(0.000)	(0.000)	(0.003)	(0.000)	(0.020)

All regressions include time, country, sector and NUTS1 dummies; p-values in brackets are based on heteroscedasticity consistent standard errors; ***, **, * denote significance at the 1, 5, and 10 % level.

Table 6: Density and spillover effects at the sectoral level in the Old and New Member States - ${\it IV}$ results

agriculture, however, we obtain a coefficient that is infeasibly large. The results on the spillover variables for old Europe tend to mirror earlier ones, with the spillover variable tending to be positive and significant for the common border measure and for high values of θ , but insignificant for smaller values of θ . For new Europe the spillover variable tends to be positive and significant for all values of θ , though hardly ever significant when the common border weighting is used.

7 Conclusions

In this paper we extend the model proposed in Ciccone and Hall (1996) and Ciccone (2002) to the industry or sectoral level. We then test the model on an extended sample of countries, which includes data on all EU member countries (with the exception of Malta) at the NUTS 2-digit level over the period 1998-2004. Our dataset allows us to consider agglomeration effects for six sectors (agriculture, manufacturing, construction, distribution services, business and public services). As an additional extension over the Ciccone paper we examine differences in the extent of agglomeration among new and old EU members.

At the aggregate level our results broadly confirm the findings of Ciccone (2002) for our extended sample of EU countries, in terms of both the significance and size of the agglomeration effect. The results are weaker once we account for endogeneity, but are still largely in line with the results of Ciccone (2002). The results at the sectoral level indicate significant agglomeration effects for five out of the six sectors, with the coefficients being similar in size to those for the total economy. For the final sector, agriculture, we often find a significant negative coefficient. Where positive the coefficients on the density variable tend to be similar in size across sectors. As with the aggregate results, the sectoral results when accounting for endogeneity tend to be weaker, though significant agglomeration effects are usually found.

While the results for the full sample are to an extent in line with existing literature, we find important differences when we split the sample into new and old Europe. In particular, we find that the effect of density on productivity in old Europe is limited, both at the aggregate and sectoral levels. For new Europe, however, we find large coefficient estimates for the effect of density on productivity, suggesting large agglomeration effects in new Europe over the period considered. As with the results for the full sample, there are few differences in the extent of agglomeration effects across the individual sectors. These large effects could be caused by clustering effects driven by foreign capital inflows in particular into bordering regions and capital

cities and concentration of economic activities in these regions. One potential implication of this result is that dense regions, such as capital cities, will have better prospects for productivity growth, while peripheral regions may not achieve catch-up. This suggests that there may be a divergence in development patterns across European regions, which the results particularly show for the regions in the new member states. However, other counteracting effects such as spillover effects across regions and a more balanced pattern of FDI inflows and investment structures may offset such an effect in the longer run.

It appears that additional analysis is needed, when looking for agglomeration effects at the sectoral or industry level, extending the Ciccone model to account for both localization and urbanization economies.

A Mathematical appendix

Average labour productivity is given by

$$\frac{Q_{isc}}{N_{isc}} = \Omega_{isc}^{\gamma_i} \left(H_{isc}^{\beta_i} \left(\frac{K_{isc}}{N_{isc}} \right)^{1-\beta_i} \right)^{\alpha_i \gamma_i} \left(\frac{N_{isc}}{A_{sc}} \right)^{\alpha_i \gamma_i - 1}$$

and demand for capital under the above mentioned assumptions is

$$K_{isc} = \frac{\alpha_i(1-\beta_i)}{r_{ic}}Q_{isc}.$$

Inserting yields

$$\begin{split} \frac{Q_{isc}}{N_{isc}} &=& \Omega_{isc}^{\gamma_i} \left(H_{isc}^{\beta_i} K_{isc}^{1-\beta_i} N_{isc}^{\beta_i-1} \right)^{\alpha_i \gamma_i} \left(\frac{N_{isc}}{A_{sc}} \right)^{\alpha_i \gamma_i - 1} \\ &=& \Omega_{isc}^{\gamma_i} \left(H_{isc}^{\beta_i} \left(\frac{\alpha_i (1-\beta_i)}{r_{ic}} Q_{isc} \right)^{1-\beta_i} N_{isc}^{\beta_i-1} \right)^{\alpha_i \gamma_i} \left(\frac{N_{isc}}{A_{sc}} \right)^{\alpha_i \gamma_i - 1} \\ &=& \Omega_{isc}^{\gamma_i} \left(H_{isc}^{\beta_i} \left(\frac{\alpha_i (1-\beta_i)}{r_{ic}} \right)^{1-\beta_i} \right)^{\alpha_i \gamma_i} \left(\frac{Q_{isc}}{N_{isc}} \right)^{(1-\beta_i)\alpha_i \gamma_i} \left(\frac{N_{isc}}{A_{sc}} \right)^{\alpha_i \gamma_i - 1} \\ \left(\frac{Q_{isc}}{N_{isc}} \right)^{1-(1-\beta_i)\alpha_i \gamma_i} &=& \Omega_{isc}^{\gamma_i} \left(H_{isc}^{\beta_i} \left(\frac{\alpha_i (1-\beta_i)}{r_{ic}} \right)^{1-\beta_i} \right)^{\alpha_i \gamma_i} \left(\frac{N_{isc}}{A_{sc}} \right)^{\alpha_i \gamma_i - 1} \end{split}$$

and thus

$$\frac{Q_{isc}}{N_{isc}} = \Omega_{isc}^{\gamma_i \frac{1}{1 - (1 - \beta_i)\alpha_i \gamma_i}} \left(H_{isc}^{\beta_i} \left(\frac{\alpha_i (1 - \beta_i)}{r_{ic}} \right)^{1 - \beta_i} \right)^{\alpha_i \gamma_i \frac{1}{1 - (1 - \beta_i)\alpha_i \gamma_i}} \left(\frac{N_{isc}}{A_{sc}} \right)^{(\alpha_i \gamma_i - 1) \frac{1}{1 - (1 - \beta_i)\alpha_i \gamma_i}}.$$

Rearranging and using the above definitions yields

$$\begin{array}{lcl} \frac{Q_{isc}}{N_{isc}} & = & \Omega_{isc}^{\frac{\gamma_{i}}{1-(1-\beta_{i})\alpha_{i}\gamma_{i}}} \left(\frac{\alpha_{i}(1-\beta_{i})}{r_{ic}}\right)^{\frac{(1-\beta_{i})\alpha_{i}\gamma_{i}}{1-(1-\beta_{i})\alpha_{i}\gamma_{i}}} \left(H_{isc}\right)^{\frac{\alpha_{i}\beta_{i}\gamma_{i}}{1-(1-\beta_{i})\alpha_{i}\gamma_{i}}} \left(\frac{N_{isc}}{A_{sc}}\right)^{\frac{\alpha_{i}\gamma_{i}-1}{1-(1-\beta_{i})\alpha_{i}\gamma_{i}}} \\ & = & \Omega_{isc}^{\omega_{i}}\Gamma_{ic} \left(H_{isc}\right)^{\theta_{i}+1} \left(\frac{N_{isc}}{A_{sc}}\right)^{\theta_{i}}. \end{array}$$

B Classifications

Industry	NACE	
code	1-digit	Description
Agriculture	A	Agriculture, hunting and forestry
	В	Fishing
Manufacturing	$^{\mathrm{C}}$	Mining
	D	Manufacturing
	Е	Electricity, gas and water supply
Construction	F	Construction
Hotels	G	Wholesale and retail trade, repair of motor vehicles, motorcycles and personal and household
		goods
	H	Hotels and restaurants
	I	Transport, storage and communication
Business services	J	Financial intermediation
	K	Real estate, renting and business activities
Public services	${ m L}$	Public administration and defence; compulsory social security
	M	Education
	N	Health and social work
	O	Other community, social and personal service activities
	P	Activities of households
	Q	Extra-territorial organisations and bodies

Table 7: Sectoral aggregates

References

- Anastassova, L. (2006). Productivity differences and agglomeration across districts of Great Britain. CERGE-EI Working Paper Series 289.
- Baldwin, R. E. and P. Martin (2004). Agglomeration and regional growth. In V. Henderson and J. Thisse (Eds.), *Handbook of Regional and Urban Economics: Cities and Geography*, Volume 4.
- Benhabib, J. and M. M. Spiegel (1994). The role of human capital in economic development: Evidence from aggregate cross-country data. *Journal of Monetary Economics* 34, 147–173.
- Brülhart, M. and N. Mathys (2008). Sectoral agglomeration economies in a panel of European regions. *Regional Science and Urban Economics forthcoming*.
- Campos, N. F. and Y. Kinoshita (2003). Why does FDI go where it goes? New evidence from the transition economies. *IMF Working Paper WP/03/228*.
- Ciccone, A. (1992). Cities and the economy: Aggregate production, urban productivity and the distribution of economic activity. Mimeo.
- Ciccone, A. (2002). Agglomeration effects in Europe. European Economic Review 46, 213–227.
- Ciccone, A. and R. Hall (1996). Productivity and the density of economic activity. *American Economic Review* 86, 54–70.
- De la Fuente, A. and X. Vives (1995). Infrastructure and education as instruments of economic policy: Evidence from Spain. *Economic Policy 20*.
- Dekle, R. and J. Eaton (1999). Agglomeration and land rents: Evidence from the prefectures. Journal of Urban Economics 46(2), 200–214.
- Ertur, C., J. L. and C. Baumont (2006). The European regional convergence process, 1980-1995: Do spatial regimes and spatial dependence matter? *International Regional Science Review* 29(1), 3–34.
- Fujita, M. and J.-F. Thisse (1996). Economics of agglomeration. *Journal of the Japanese and International Economics* 10, 339–378.

- Henderson, V. J. (1974). The sizes and types of cities. American Economic Review 64(4), 640–656.
- Hohenberg, P. and L. Lees (1985). *The Making of Urban Europe 1000-1950*. Cambridge, MA: Harvard University Press.
- Kim, S.-H., T. S. Pickton, and S. Gerking (2003). Foreign direct investment: Agglomeration economies and returns to promotion expenditures. *The Review of Regional Studies* 33(1), 61–72.
- Krugman, P. R. (1991). Increasing returns and economic geography. *Journal of Political Economy* 99, 483–499.
- Krugman, P. R. and A. J. Venables (1995). Globalization and the inequality of nations. *Quarterly Journal of Economics* 60, 857–880.
- Marshall, A. (1920). Principles of Economics. London: MacMillan.
- Mills, E. S. (1967). An aggregative model of resource allocation in a metropolitan area. *American Economic Review, Papers and Proceedings* 57(2), 197–210.
- Pelegrín, À. and C. Bolancé (2008). Regional foreign direct investment in manufacturing: Do agglomeration economies matter? *Regional Studies* 42(4), 505–522.
- Pritchett, L. (2001). Where has all the education gone? World Bank Economic Review 15(3), 367–391.
- Rice, P., A. Venables, and E. Patacchini (2006). Spatial determinants of productivity: Analysis for the regions of Great Britain. *Regional Science and Urban Economics* 36(6), 727–752.
- Rosenthal, S. S. and W. C. Strange (2003). Geography, industrial organization and agglomeration. *Review of Economics and Statistics* 85(2), 377–393.
- Rosenthal, S. S. and W. C. Strange (2004). Evidence on the nature and sources of agglomeration economies. In Henderson, V. and J.F. Thisse (eds), Handook of Regional and Urban Economics, Volume 4. North Holland, 85(2), 377–393.

Short list of the most recent wiiw publications (as of September 2008)

For current updates and summaries see also wiiw's website at www.wiiw.ac.at

Sectoral Productivity, Density and Agglomeration in the Wider Europe

by Neil Foster and Robert Stehrer

wiiw Working Papers, No. 47, September 2008 25 pages including 7 Tables hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

The Vienna Institute Monthly Report 8-9/08

edited by Leon Podkaminer

- Economic consequences of the Georgian-Russian conflict
- EU Structural Funds in Central and East European countries
- Czechoslovak economic reforms in the 1960s
- Statistics: wiiw Database on Foreign Direct Investment in Central, East and Southeast Europe, 2000-2007

wiiw, August-September 2008 36 pages including 11 Tables and 8 Figures (exclusively for subscribers to the wiiw Service Package)

Skill Diffusion by Temporary Migration? Returns to Western European Work Experience in Central and East European Countries

by Anna lara

wiiw Working Papers, No. 46, July 2008 43 pages including 14 Tables hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

The Big Boom Is Over, but Growth Remains Strong and Inflation Calms Down

by Leon Podkaminer, Josef Pöschl et al.

wiiw Current Analyses and Forecasts. Economic Prospects for Central, East and Southeast Europe, No. 2, July 2008

160 pages including 50 Tables and 16 Figures hardcopy: EUR 70.00 (PDF: EUR 65.00)

The Vienna Institute Monthly Report 7/08

edited by Leon Podkaminer

- Hungarian agriculture starting the fifth year within the European Union
- Migration from and to Hungary
- The international role of the euro: prospects of dethroning the dollar as the leading international currency still fairly remote
- Statistics: Selected monthly data on the economic situation in Central, East and Southeast Europe, 2002-2008

wiiw, July 2008

36 pages including 8 Tables and 17 Figures (exclusively for subscribers to the wiiw Service Package)

wiiw Database on Foreign Direct Investment in Central, East and Southeast Europe, 2008: Decline to Follow Uneven FDI Inflow Growth

by Gábor Hunya. Database and layout by Monika Schwarzhappel.

wiiw, Vienna, June 2008

114 pages including 84 Tables and 4 Figures

hardcopy: EUR 70.00 (PDF: EUR 65.00), CD-ROM (including hardcopy): EUR 145.00

The Vienna Institute Monthly Report 6/08

edited by Leon Podkaminer

- Converging/diverging European regions
- The new EU members' potential for trade in services
- · Market economy needs to run budgetary deficits
- Statistics: Selected monthly data on the economic situation in Central, East and Southeast Europe, 2007-2008

wiiw, June 2008

32 pages including 16 Tables and 5 Figures

(exclusively for subscribers to the wiiw Service Package)

MOEL: Wachstumsvorsprung gegenüber Westeuropa bleibt erhalten

by Vasily Astrov

wiiw Research Papers in German language, May 2008

(reprinted from: WIFO-Monatsberichte, Vol. 81, No. 5, May 2008)

16 pages including 8 Tables and 3 Figures

hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Facing the Monster 'Juste retour': On the Net Financial Position of Member States vis-à-vis the EU Budget and a Proposal for Reform

by Sándor Richter

wiiw Research Reports, No. 348, May 2008 109 pages including 54 Tables and 2 Figures hardcopy: EUR 22.00 (PDF: EUR 20.00)

The Vienna Institute Monthly Report 5/08

edited by Leon Podkaminer

- Terms of trade effects and Russian economic growth
- Poland: the national stock exchange game
- Some reflections on the current US financial markets meltdown
- Statistics: Regional disparities and economic developments in the enlarged EU wiiw, May 2008

36 pages including 3 Tables, 9 Figures and 8 Maps (exclusively for subscribers to the wiiw Service Package)

What Is in the Name? Risk Assessment of Macedonia

by Vladimir Gligorov

wiiw Research Reports, No. 347, April 2008 34 pages including 15 Tables and 6 Figures hardcopy: EUR 22.00 (PDF: EUR 20.00)

Economic and Trade Policy Impacts of Sustained High Oil Prices

by Gábor Pellényi and Edward Christie (eds), Judit Barta, Miklós Hegedüs, Mario Holzner, András Oszlay and Magdolna Sass

> wiiw Research Reports, No. 346, April 2008 69 pages including 4 Tables and 19 Figures hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

The Vienna Institute Monthly Report 4/08

edited by Leon Podkaminer

- Skills and the competitiveness of EU manufacturing industries
- · Selected economic impacts of higher oil prices
- Selected monthly data on the economic situation in ten transition countries, 2005-2008 wiiw, April 2008

26 pages including 8 Tables and 15 Figures (exclusively for subscribers to the wiiw Service Package)

The Vienna Institute Monthly Report 3/08

edited by Leon Podkaminer

- Armenia and Azerbaijan: recent economic developments and policy challenges
- Agricultural support: consequences of an eventual EU accession of the Balkan countries
- Globalization and inflation: impacts unlikely to be large and permanent
- Selected monthly data on the economic situation in ten transition countries, 2002-2008 wiiw, March 2008

32 pages including 6 Tables and 12 Figures (exclusively for subscribers to the wiiw Service Package)

Weathering the Global Storm, yet Rising Costs and Labour Shortages May Dampen Domestic Growth

by Peter Havlik, Mario Holzner et al.

wiiw Current Analyses and Forecasts. Economic Prospects for Central, East and Southeast Europe, No. 1, February 2008 $\,$

151 pages including 51 Tables and 20 Figures hardcopy: EUR 70.00 (PDF: EUR 65.00)

wiiw Service Package

The Vienna Institute offers to firms and institutions interested in unbiased and up-to-date information on Central, East and Southeast European markets a package of exclusive services and preferential access to its publications and research findings, on the basis of a subscription at an annual fee of EUR 2,000.

This subscription fee entitles to the following package of **Special Services**:

- A free invitation to the Vienna Institute's Spring Seminar, a whole-day event at the end of March, devoted to compelling topics in the economic transformation of the Central and East European region (for subscribers to the wiiw Service Package only).
- Copies of, or online access to, *The Vienna Institute Monthly Report*, a periodical consisting of timely articles summarizing and interpreting the latest economic developments in Central and Eastern Europe and the former Soviet Union. The statistical annex to each *Monthly Report* contains, alternately, country-specific tables or graphs with monthly key economic indicators, economic forecasts, the latest data from the wiiw Industrial Database and excerpts from the wiiw FDI Database. This periodical is not for sale, it can only be obtained in the framework of the wiiw Service Package.
- Free copies of the Institute's Research Reports (including Reprints), Current Analyses and Forecasts, Country Profiles and Statistical Reports.
- A free copy of the wiiw Handbook of Statistics (published in October/November each year and containing more than 400 tables and graphs on the economies of Albania, Bosnia and Herzegovina, Bulgaria, Croatia, the Czech Republic, Estonia, Hungary, Latvia, Lithuania, Macedonia, Montenegro, Poland, Romania, Russia, Serbia, the Slovak Republic, Slovenia and Ukraine)
- Free online access to the wiiw Monthly Database, containing more than 1200 leading indicators monitoring the latest key economic developments in ten Central and East European countries.
- Consulting. The Vienna Institute is pleased to advise subscribers on questions concerning
 the East European economies or East-West economic relations if the required background
 research has already been undertaken by the Institute. We regret we have to charge extra
 for ad hoc research.
- Free access to the Institute's specialized economics library and documentation facilities.

Subscribers who wish to purchase wiiw data sets **on CD-ROM** or special publications not included in the wiiw Service Package are granted considerable **price reductions**.

For detailed information about the wiiw Service Package please visit wiiw's website at www.wiiw.ac.at

To The Vienna Institute for International Economic Studies Oppolzergasse 6 A-1010 Vienna

- O Please forward more detailed information about the Vienna Institute's Service Package
- O Please forward a complete list of the Vienna Institute's publications to the following address

Please enter me for

O 1 yearly subscription of *Research Reports* (including *Reprints*) at a price of EUR 225.00 (within Austria), EUR 250.00 (Europe) and EUR 265.00 (overseas) respectively

Plea	ase forward		
0	the following issue of Research Reports		
0	the following issue of Current Analyses and	d Forecasts	
0	the following issue of Country Profiles		
0	the following issue of Working Papers		
0	the following issue of Statistical Reports		
0	the following issue of Research Papers in 0	German language	
0	the following issue of wilw Database on Fo	reign Direct Investment	
0	the following issue of wilw Handbook of Sta	atistics	
Nar	ne		
۸۵۵	Iress		
Auc	11000		
		F	
reie	ephone	Fax	E-mail
Date	е		Signature

Herausgeber, Verleger, Eigentümer und Hersteller:

 $Verein\ \hbox{,} Wiener\ Institut\ f\"ur\ Internationale\ Wirtschaftsvergleiche"\ (wiiw),$

Wien 1, Oppolzergasse 6

Postanschrift: A-1010 Wien, Oppolzergasse 6, Tel: [+431] 533 66 10, Telefax: [+431] 533 66 10 50

Internet Homepage: www.wiiw.ac.at

Nachdruck nur auszugsweise und mit genauer Quellenangabe gestattet.

P.b.b. Verlagspostamt 1010 Wien